Author's response to reviews

Title: Efficacy of manual therapy to avoid progression of adolescent idiopathic scoliosis: A systematic review

Authors:

Michele Romano (michele.romano@isico.it)
Stefano Negrini (stefano.negrini@isico.it)

Version: 3 Date: 11 January 2008

Author's response to reviews: see over
We thank the reviewers for their important input.

Changes to the manuscript

We changed the title according to Major compulsory revision required.

We do not agree with the idea of calling "traditional" the “official” medicine, because the former imply that there are other approaches to medicine at the same level. Also official is not such a positive judgement of physician's approach, but nevertheless must be recognized that medicine is growing because of its EBM approach, while alternative/complimentar medicines will grow only when they will accept the methods now used by official medicine. We then maintained “official”.

Last point: we inserted sub-titles to let readers understand better the contents and avoid any confusion.